Author's response to reviews

Title: Diagnostic accuracy of semi-quantitative and quantitative culture techniques for the diagnosis of catheter-related infections in newborns and molecular typing of isolated microorganisms

Authors:

Danilo FM Riboli (danilo.riboli@ibb.unesp.br)
João C Lyra (jc.lyra@uol.com.br)
Eliane P Silva (eliane_bio10@yahoo.com.br)
Luisa L Valadão (luisaleit@gmail.com)
Maria R Bentlin (mbentlin@fmb.unesp.br)
José E Corrente (jecorren@ibb.unesp.br)
Lígia MSS Rugolo (ligiasr@fmb.unesp.br)
Maria LRS Cunha (cunhamlr@ibb.unesp.br)

Version: 5  Date: 2 April 2014

Author's response to reviews: see over
April 02, 2014

Sheryl Ramos
Journal Editorial Office
BioMed Central
Ms. Ref. No.: 6724602341044985

Dear Sheryl Ramos,

Enclosed please find the revised version of our manuscript “Diagnostic accuracy of semi-quantitative and quantitative culture techniques for the diagnosis of catheter-related infections in newborns and molecular typing of isolated microorganisms”, which was modified according to the suggestions of the editors. We thank the editors by decision to accept the manuscript and for the comments that helped improve the manuscript.

The changes suggested by editors are highlighted in bold in the text. We remain at your disposal for further clarification.

Sincerely,

Danilo Flávio Moraes Riboli
Departamento de Microbiologia e Imunologia
Instituto de Biociências, UNESP, Botucatu, SP, Brazil

Please include the name of the ethics committee who approved your study in the manuscript.

The text was changed accordingly.
<table>
<thead>
<tr>
<th>Page</th>
<th>As it was</th>
<th>Modification</th>
</tr>
</thead>
<tbody>
<tr>
<td>Materials and methods 06</td>
<td>The strains had been isolated between November 2005 and February 2011 and were stored in a freezer at -70°C. The present study was approved by the institutional Ethics Committee (Protocol 3522-2010).</td>
<td>The strains had been isolated between November 2005 and February 2011 and were stored in a freezer at -70°C. The present study was approved by Committee for Ethics in Research (“Comitê de Ética em Pesquisa” from “Faculdade de Medicina de Botucatu”, Botucatu, São Paulo State, Brazil - Protocol 3522-2010).</td>
</tr>
</tbody>
</table>